Author's response to reviews

Title: Impact of naturally occurring amino acid variations on the detection of HIV-1 p24 in diagnostic antigen tests

Authors:

Beatrice N Vetter (vetter.beatrice@virology.uzh.ch)
Vanessa Orlowski (orlowski.vanessa@virology.uzh.ch)
Christoph Niederhauser (christoph.niederhauser@bsd-be.ch)
Louise Walter (louise.walter@swissmedic.ch)
Jörg Schüpbach (schupbach.jorg@virology.uzh.ch)

Version: 4  Date: 14 September 2015

Author's response to reviews: see over
Revised manuscript: Impact of naturally occurring amino acid variations on the detection of HIV-1 p24 in diagnostic antigen tests

Dear Editor,

According to the editorial request made on the 3rd of September, we have included a reference to the Swiss legislation in the manuscript paragraph addressing ethical approval (lines 133-137).

With best regards,

Dr. Beatrice Vetter